Author’s response to reviews

Title: Design of the Growth hormone deficiency and Efficacy of Treatment (GET) score and non-interventional proof of concept study

Authors:

Peter Kann (kannp@med.uni-marburg.de)
Simona Bergmann (simona.bergmann@med.uni-marburg.de)
Martin Bidlingmaier (Martin.Bidlingmaier@med.uni-muenchen.de)
Christina Dimopoulou (chdimo@hotmail.com)
Birgitte Pedersen (BITP@novonordisk.com)
Günter Stalla (stalla@mpipsyk1.mpg.de)
Matthias Weber (mmweber@uni-mainz.de)
Stefanie Meckes-Ferber (STMF@novonordisk.com)

Version: 3 Date: 29 Jan 2018

Author’s response to reviews:

Dear Editorial Office,

Thank you for the opportunity to address those final comments. Responses to individual comments follow:

1) Informed consent is referred to in line 183, as well as in the 'Ethics and consent to participate' section (lines 385-390).

2) Individual contributions for each author have been listed in full.

Kind regards